Author's response to reviews

Title: Ectopic thyroid in an adrenal mass: A case report

Authors:

Jun Hagiuda (junhagi@hotmail.com)
Isao Kuroda (veq01625@nifty.ne.jp)
Takuii Tsukamoto (007tt@saitama-med.ac.jp)
Munehisa Ueno (camtmune@saitama-med.ac.jp)
Chizuko Yokota (cyokota@saitama-med.ac.jp)
Takanori Hirose (thirose@saitama-med.ac.jp)
Nobuhiro Deguchi (ndeguchi@saitama-med.ac.jp)

Version: 2 Date: 24 June 2006

Author's response to reviews:

Dear Editor:

Thank you very much for your comments to our manuscript (MS: 4336947861010764) and recommendations for revisions. We have responded to each of them and made a few changes in the text. Please note that we added a few sentences to the last paragraph of case presentation, made a new photograph for Figure 3b, rearranged the reference list, asked an English native speaker to look through our manuscript and made a few minor linguistic corrections, and changed and rephrased the sentence of the second paragraph of the conclusion.

We will be pleased that you review our revised paper again and accept it for publication in BMC Urology. Though we have asked an English native speaker to correct our manuscript, I would like to request for a copy-editing service if it is necessary.

Sincerely yours,

Jun Hagiuda, MD
Saitama Medical University
Department of Urology
38 Morohongo, Moroyamamachi, Iruma
Saitama 3500495, Japan
Telephone: 81-492-76-1243
Fax: 81-492-95-7338
E-mail: junhagi@hotmail.com

To Reviewer 1

Manuscript Re:4336947861010764
Ectopic thyroid in an adrenal mass: A case report

I really thank you for your comments, each of which seems to be very reasonable. We have responded to them and made a few corrections as following.

#1 We added the next sentence to the last paragraph of case presentation; Although this patient has been followed up for 2 years after the operation, laboratory study and computed tomography failed to show any new lesions.

#2 We made a new photograph for Figure 3b. Please, find enclosed a new picture.

To Reviewer 2

Manuscript Re:4336947861010764
Ectopic thyroid in an adrenal mass: A case report
I really thank you for your comments, each of which seems to be very reasonable. We have responded to them and made a few corrections as following.

#1 We rearranged the reference list.

#2 We asked an English native speaker to look through our manuscript and made a few minor linguistic corrections.

#3 We changed and rephrased the sentence of the second paragraph of the conclusion.

#4 We made a new photograph for Figure 3b. Please, find enclosed a new picture.